Massachusetts Institute of Technology Handout 14 6.852: Distributed Algorithms Prof. Nancy Lynch October 23, 2003

Problem Set 4, Part b

Due: Thursday, October 30, 2003

Reading:

Distributed Algorithms, Sections 16.5 and 16.6 Chapter 18 Mattern paper Chapter 19 Chapter 9 (skim) Start Chapter 10 (Sections 10.1-10.5)

Reading for next week:

Finish Chapter 10 (Sections 10.6-10.9). Chapter 11 (skip 11.4) Chapter 12

Problems:

1. Exercise 18.4.

2. Exercise 18.10.

3. Exercise 18.17.

For (a), just state the theorem, no proof.

For (b), just give the key ideas, not a detailed formal proof.